\documentclass{jarticle}

\def\noop#1{}
\def\JLaTeX{J\LaTeX}
\def\JTeX{J\TeX}

\oddsidemargin0mm
\evensidemargin0mm
\topmargin-15mm
\textwidth162mm
\textheight245mm

\begin{document}
\section{$B2$J8(B}

jbibtex$B$N%F%9%H(B\cite{article-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{article-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{article-crossref}$B!#(B
jbibtex$B$N%F%9%H(B\cite{whole-journal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{inbook-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{inbook-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{inbook-crossref}$B!#(B
jbibtex$B$N%F%9%H(B\cite{book-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{book-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{book-crossref}$B!#(B
jbibtex$B$N%F%9%H(B\cite{whole-set}$B!#(B
jbibtex$B$N%F%9%H(B\cite{booklet-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{booklet-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{incollection-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{incollection-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{incollection-crossref}$B!#(B
jbibtex$B$N%F%9%H(B\cite{whole-collection}$B!#(B
jbibtex$B$N%F%9%H(B\cite{manual-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{manual-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{mastersthesis-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{mastersthesis-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{misc-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{misc-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{inproceedings-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{inproceedings-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{inproceedings-crossref}$B!#(B
jbibtex$B$N%F%9%H(B\cite{proceedings-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{proceedings-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{whole-proceedings}$B!#(B
jbibtex$B$N%F%9%H(B\cite{phdthesis-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{phdthesis-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{techreport-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{techreport-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{unpublished-minimal}$B!#(B
jbibtex$B$N%F%9%H(B\cite{unpublished-full}$B!#(B
jbibtex$B$N%F%9%H(B\cite{random-note-crossref}$B!#(B

\section{$BF|K\8l$N%9%?%$%k$N%F%9%HMQ$N%G!<%?72(B}

jbibtex$B$N%F%9%H(B\cite{inbook-full-j}$B!#(B
jbibtex$B$N%F%9%H(B\cite{incol-full-j-1}$B!#(B
jbibtex$B$N%F%9%H(B\cite{article-crossref-j}$B!#(B
jbibtex$B$N%F%9%H(B\cite{article-crossref-jj}$B!#(B
jbibtex$B$N%F%9%H(B\cite{whole-journal-j}$B!#(B
jbibtex$B$N%F%9%H(B\cite{inbook-crossref-j}$B!#(B
jbibtex$B$N%F%9%H(B\cite{whole-set-j}$B!#(B
jbibtex$B$N%F%9%H(B\cite{inbook-crossref-j-1}$B!#(B
jbibtex$B$N%F%9%H(B\cite{cvs}$B!#(B
jbibtex$B$N%F%9%H(B\cite{ha}$B!#(B
jbibtex$B$N%F%9%H(B\cite{costa}$B!#(B
jbibtex$B$N%F%9%H(B\cite{mcclella}$B!#(B
jbibtex$B$N%F%9%H(B\cite{dug}$B!#(B
jbibtex$B$N%F%9%H(B\cite{sakawa}$B!#(B
jbibtex$B$N%F%9%H(B\cite{ssl}$B!#(B
jbibtex$B$N%F%9%H(B\cite{newman}$B!#(B
jbibtex$B$N%F%9%H(B\cite{Rich}$B!#(B
jbibtex$B$N%F%9%H(B\cite{goto}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B0k:j(B}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B@FF#(B}$B!#(B
jbibtex$B$N%F%9%H(B\cite{sym}$B!#(B
jbibtex$B$N%F%9%H(B\cite{eda}$B!#(B
jbibtex$B$N%F%9%H(B\cite{dss}$B!#(B
jbibtex$B$N%F%9%H(B\cite{cm}$B!#(B
jbibtex$B$N%F%9%H(B\cite{reduce}$B!#(B
jbibtex$B$N%F%9%H(B\cite{fp}$B!#(B
jbibtex$B$N%F%9%H(B\cite{la}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B$"$U$lL5$7(B}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B%@%`(B}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B?ML>I=5-(B}$B!#(B
jbibtex$B$N%F%9%H(B\cite{EUC$BF|K\8l(BTeX}$B!#(B
jbibtex$B$N%F%9%H(B\cite{multi}$B!#(B
jbibtex$B$N%F%9%H(B\cite{marumoji}$B!#(B
jbibtex$B$N%F%9%H(B\cite{maru}$B!#(B

\section{yomi $B$KJ?2>L>$r;H$&Nc(B}

jbibtex$B$N%F%9%H(B\cite{goto-h}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B0k:j(B-h}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B@FF#(B-h}$B!#(B
jbibtex$B$N%F%9%H(B\cite{multi-h}$B!#(B

\section{$B%=%U%H%&%'%"2J3X2qMQ$N%F%9%H%G!<%?(B}

jbibtex$B$N%F%9%H(B\cite{ama}$B!#(B
jbibtex$B$N%F%9%H(B\cite{Arv}$B!#(B
jbibtex$B$N%F%9%H(B\cite{cha}$B!#(B
jbibtex$B$N%F%9%H(B\cite{dav}$B!#(B
jbibtex$B$N%F%9%H(B\cite{den}$B!#(B
jbibtex$B$N%F%9%H(B\cite{fis}$B!#(B
jbibtex$B$N%F%9%H(B\cite{gajski}$B!#(B
jbibtex$B$N%F%9%H(B\cite{suna1}$B!#(B
jbibtex$B$N%F%9%H(B\cite{suna2}$B!#(B
jbibtex$B$N%F%9%H(B\cite{toko1}$B!#(B
jbibtex$B$N%F%9%H(B\cite{toko2}$B!#(B
jbibtex$B$N%F%9%H(B\cite{suna86}$B!#(B

\section{$B?M9)CNG=3X2q;oMQ%F%9%H%G!<%?(B}

jbibtex$B$N%F%9%H(B\cite{onda}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B6aF#(B}$B!#(B
jbibtex$B$N%F%9%H(B\cite{JSAI-1}$B!#(B
jbibtex$B$N%F%9%H(B\cite{grosz}$B!#(B

\section{$B2~9T0LCV$N%F%9%H(B}

jbibtex$B$N%F%9%H(B\cite{$B2~9T0LCV(B2a}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B2~9T0LCV(B2b}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B2~9T0LCV(B3a}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B2~9T0LCV(B3b}$B!#(B
jbibtex$B$N%F%9%H(B\cite{$B2~9T0LCV(B3c}$B!#(B


%\bibliographystyle{jplain}
\bibliographystyle{jalpha}
%\bibliographystyle{jabbrv}
%\bibliographystyle{junsrt}
%\bibliographystyle{jname}

%\bibliographystyle{tipsj}%% $B>pJs=hM}3X2qO@J8;o(B
%\bibliographystyle{jipsj}%% $B>pJs=hM}3X2q2$J8O@J8;o(B
%\bibliographystyle{tieice}%% $BEE;R>pJsDL?.3X2qO@J8;o(B
%\bibliographystyle{jorsj}%% $BF|K\%*%Z%l!<%7%g%s%:%j%5!<%A3X2qO@J8;o(B

\bibliography{jxampl,linebreak}

\end{document}
